
\begin{figure}
\begin{center}
\begin{tabular}{ccccc}
  & {Confinement $\beta=5.2$}  & \hspace*{8mm}   & &
  {Deconfinement $\beta=5.4$} \\
  \vspace*{0mm}
  & $ma=0.05$ &&& $ma=0.05$ \\[2mm]
  \multicolumn{2}{c}{\epsfxsize=4.75cm\epsffile{ps5.2.005.eps}} &&
  \multicolumn{2}{c}{\epsfxsize=4.75cm\epsffile{ps5.4.005.eps}}
\end{tabular}
\end{center}
\vspace*{-3mm}
\caption{Nearest-neighbor spacing distribution $P(s)$ on a $6^3 
  \times 4$ lattice in full QCD (histograms) compared with the
  random matrix result (solid lines). There are no changes in $P(s)$
  across the deconfinement phase transition.}
\vspace*{-3mm}
\label{fintemp}
\end{figure}

